Author's response to reviews

Title: Excision of sympathetic ganglia and the rami communicantes with histologic confirmation offers better early and late outcomes in Video assisted thoracoscopic sympathectomy.

Authors:

Sridhar Rathinam (srathinam@rcsed.ac.uk)
Prakash Nanjaiah (drprks@yahoo.com)
Sivakumar Sivalingam (sivkuma71@yahoo.com)
Pala B Rajesh (pala.rajesh@heartofengland.nhs.uk)

Version: 6 Date: 13 August 2008

Author's response to reviews:

Dear Sir,

a few typographical errors were amended.
1. Adding a colon in page 5 after grade 2
2. page 6 Medical notes review: addition of s
3. Results in the study time frame Addition of word study
4. Follow up: deletion of of before 67.8 months
5. page 9 In our series instead of study
6. Page 9 comparable to the 6.5% reported in literature

Thank you
Your sincerely

Sri Rathinam